Author's response to reviews

Title: The effect of transcutaneous electrical acupoint stimulation on pregnancy rates in women undergoing in vitro fertilization: a study protocol for a randomized controlled trial

Authors:

Cui Hong Zheng (chzheng2008@qq.com)
Juan Zhang (1264298243@qq.com)
Jing Wu (wujingrlh@163.com)
Ming Min Zhang (mmzhang@tjh.tjmu.edu.cn)

Version: 3 Date: 17 April 2014

Author's response to reviews:

1. Please include the date your study was registered with your trial registration number in the Abstract.

   Answer: We have included the date in the new version.
   Trial registration: ClinicalTrials.gov: ID: NCT01608048 (05/24/2012).

2. Please include a list of abbreviations used in the manuscript and their meanings after your Trial Status.

   Answer: We have added this part in the new version.

   Abbreviations

3. Please rename your statement of funding 'Acknowledgements'.

   Answer: We have renamed the statement of funding 'Acknowledgements in the new version.

4. Please include a figure title and legend section after the reference list. The figures should not be included in the main body of the manuscript.

   Answer: We have added the figure title after the reference list.